Author’s response to reviews

Title: Development and validation of a Diabetes Risk Score for screening undiagnosed diabetes in Sri Lanka (SLDRISK)

Authors:

Prasad Katulanda (prasad.katulanda@yahoo.com)
Nathan Hill (nathan.hill@phc.ox.ac.uk)
Irene Stratton (ims_55@yahoo.co.uk)
Rezvi Sheriff (rezvi.sheriff@gmail.com)
Sawanawadu Neomal De Silva (neomalde@gmail.com)
David Matthews (david.matthews@hmc.ox.ac.uk)

Version: 3 Date: 13 Jul 2016

Author’s response to reviews:

1. We have recently changed our guidance for authors and we now require a Declarations section.

Please provide a COMPLETE Declarations section before the references, including an Availability of data and materials section (we appreciate that you may have supplied some of the information in the body of the manuscript).

Please move the Abbrevations from the title page to this section - Done

- Please include a statement on ethics approval and consent to participate.

Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate - Done
- Consent for publication – Not applicable as our paper doesn't contain any individual data.

- Availability of data and materials – Done

- Funding – required

  We appreciate that a statement providing funding information may already be present. Please move/copy the statement to the Funding sub-section instead, and include any additional relevant information - Done

- Competing Interests – (already provided)

- Authors’ contributions – amendment required

  Currently, the contributions of SDNDS do not automatically qualify them for authorship. Please provide more information on their contributions, or remove their name from the list of authors - More information provided on contribution

- Acknowledgements – (already provided)

- Authors’ Information (optional) – (already provided)

2. Please make sure the abstract contains the following headings only: Background, Methods, Results, Conclusions. Please see our Submission guidelines for more information (http://bmcendocrdisord.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article).
3. Please change the "Introduction" heading to "Background" - Done

4. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance - Done

5. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Currently, the supplementary file and Table 3 are not cited - These were cited

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables and figures should also be clean versions - Done